

## Specific Comments To Authors

**Comment 1:** The authors describe in detail some of the effects of statins and note their association with some pancreatic diseases. The incidence of idiopathic chronic pancreatitis is relatively high in the author's locality, and the author's study has good clinical value. The authors found that statins may be useful in the treatment of idiopathic chronic pancreatitis. We look forward to the results of the authors' prospective clinical study.

**Response to the comment 1:** In view of preliminary data authors are very optimistic about the role of simvastatin with or without N-Acetylcysteine in idiopathic chronic pancreatitis.